Author's response to reviews

Title: Implementation of evidence-based practice across medical, nursing, pharmacological and allied health professionals in a nationwide hospital setting

Authors:

Yi-Hao Weng (yihaoweng@adm.cgmh.org.tw)
Ken N Kuo (kennank@aol.com)
Chun-Yuh Yang (chunyuh@kmu.edu.tw)
Heng-Lien Lo (danielnote@gmail.com)
Chiehfeng Chen (clifchen@tmu.edu.tw)
Ya-Wen Chiu (bettychiu@tmu.edu.tw)

Version: 5 Date: 17 September 2013

Author's response to reviews: see over
Dear Editorial Team,

We accept all changes you have made in the text. In addition, we have changed some parts of the manuscript, including those that you have pointed out. Please refer to the following explanations.

1. We have added a citation for validity and reliability (page 7).
2. We have added two more funding resources in the acknowledgement (page 15).
3. We have used plural nouns in Figure 1.
4. We have made the all professions as singular noun in Table 2 (page 23).

Please do not hesitate to contact us if you have any further questions.

Best regards,

Ya-Wen Chiu, PhD
Associate Professor
Master Program in Global Health and Development
College of Public Health and Nutrition
Taipei Medical University, Taiwan